Author's response to reviews

Title: Socio-economic and demographic risk factors on child drowning in Bangladesh

Authors:

Mosharaf Md. Hossain (m_population@yahoo.com)
Kulanthayan Dr. K.C. Mani (kulan@upm.edu.my)
Sherina Dr. Mohd Sidik (sherina@upm.edu.my)
Hayati Dr. KS (hayatik@upm.edu.my)
AKM Fazlur Dr. Rahman (fazlur@ciprb.org)

Version: 5 Date: 9 July 2015

Author's response to reviews:

Dear Editor

Thanks for your information and also many thanks to reviews. This article already English checked by John Dilling, Professional Editor from UK.

Line 52........Done
Line 66-67.......Done
Line 70..........Done
Line 71.........Done
Line 74..............Done
Line 75-77.........Done
Line 98-100.......Done
Line 106-107.......Done
Line 112-113.......Done
Line 117.........Done
Line 154.........Done
Line 173-174.......Done
Line 177-179.......Done
Line 181..............Done
Line 184..............Done
Line 191-193.......Done
Line 197-198.......Done
Line 201-202.......Done
Line 204-206...........Done
Line 214-218........Done
Line 224-226..................Done
Line 229......................Done
Line 240-241...............Done
Line 244-249................Done
Line 253.....................Done
Line 256-259...............Done
Line 267.....................Done

Thanks
Mosharaf